# Conflicts of interest on corporate boards: The effect of creditor-directors on acquisitions

Jens Hilscher Elif Şişli-Ciamarra

August 2011

#### Abstract

This paper investigates the effects on acquisitions of creditor-director presence on corporate boards. Using a hand-collected dataset for boards of large U.S. corporations, we find that companies with creditor-directors are more likely to engage in acquisitions with attributes that are unfavorable to shareholders and favorable to creditors (more diversifying and fewer cash-financed acquisitions). Consistent with these patterns, acquisition announcements are associated with lower shareholder value, higher creditor value, and lower overall firm value when a creditor is present. These results support the hypothesis that conflicts of interest between shareholders and creditors result in value-destroying acquisitions. In addition, commercial bankers with no lending relationship are not affected by conflicts of interest. Where appropriate, our estimation strategy takes into account that there may be self selection of bankers onto corporate boards.

JEL Classification: G21, G34

Keywords: shareholder-creditor conflicts, acquisitions, board of directors, bankers on boards, corporate governance, credit market reaction

<sup>\*</sup> Both authors are at the International Business School, Brandeis University, 415 South Street, Waltham MA 02453. E-mail: hilscher@brandeis.edu, esisli@brandeis.edu. We are indebted to William Greene, Lawrence White, David Yermack, and Bernard Yeung for their valuable guidance and suggestions. We would like to thank Franklin Allen, Abigail Hornstein, Blake LeBaron, Shawn Mobbs, Carol Osler, Kartik Raman, Monica Singhal, and participants at the NYU Corporate Governance Workshop (2007), Boston Area Finance Symposium (2009), FMA Annual Meetings (2009), Brandeis Brownbag Seminar (2010), Wesleyan University Research Seminar (2010), and UMass Amherst Research Seminar (2011) for helpful comments and discussions. We acknowledge financial support from Brandeis International Business School. We also thank Georgiana Buturoiu, Jeremy Karp, Xia Meng, Huong Nguyen, Elona Rika, Saoni Sen, Ly Tran, and especially Nikolaos Petros Petropoulos for assistance with data collection, and Manjola Tase for research assistance. All errors and omissions are our own.

#### 1. Introduction

Acquisitions are among the most important corporate decisions and may result in significant wealth transfers from shareholders to creditors (Galai and Masulis, 1976). The main corporate governance mechanism to prevent such wealth transfers is the oversight function performed by the acquiring company's board of directors. Creditors are often represented on corporate boards: 11.4 percent of S&P 500 companies had a creditor on their boards between 2002 and 2007. Like any other board member, a creditor-director has a fiduciary duty to disapprove of acquisitions that do not increase shareholder value. Nonetheless, conflicts of interest between shareholders and creditors may cause creditors to approve acquisitions that add value to creditors but reduce shareholder value (Jensen and Meckling, 1976).

An alternative possibility is that creditors perform important governance functions: Due to their lending relationships, creditors may have access to information that other board members may not have. The informational advantage of creditors may facilitate monitoring and enable the board to screen out value-destroying projects (Boyd and Prescott, 1986; Diamond 1991).

In this paper we present evidence that creditor presence on a company's board of directors is associated with acquisitions that are unfavorable for shareholders but favorable for creditors. We find a negative equity market and a positive credit market reaction to acquisition announcements. The market reactions reflect that creditor presence is associated with types of acquisitions that may add value to the creditors and reduce shareholder value: more diversifying and fewer acquisitions that are paid entirely by cash. These findings are in support of the conflicts of interest hypothesis.

We start by building a hand-collected data set of director characteristics for the S&P 500 non-financial firms between 2002 and 2007. We identify creditors as those board members whose primary employer is a commercial bank with an outstanding lending relationship with the

company. We then gather data on acquisitions that were announced and completed by our sample firms between 2002 and 2007 and analyze the effect of creditor presence on acquisition activity. Specifically, we consider acquisition intensity, deal attributes, and equity and credit market reactions to acquisition announcements.

In our empirical analysis we pay special attention to the possibility of self-selection of creditors onto corporate boards. Previous studies show that companies with and without creditors on their boards have significantly different financial characteristics (Booth and Deli, 1999; Kroszner and Strahan, 2001; Byrd and Mizruchi, 2005; Rumble and Santos, 2006; Güner et al., 2008; Ferreira and Matos, 2011; Şişli-Ciamarra, 2011). Therefore, we may identify differences in firms with and without a creditor-director either because there is an effect of creditor presence on acquisition quality or because firms with characteristics that make them more likely to have creditors on their boards are also more likely to engage in more or less favorable acquisitions. To address the possibility of self-selection, we explicitly model creditor selection within an average treatment effects framework (Heckman, 1978): We test if there is evidence for selection, and, when appropriate, correct for it by jointly estimating the selection of bankers onto boards and the effect of creditors on acquisitions.

We first study if creditor presence is associated with a higher propensity to acquire, or with types of acquisitions that have been shown to reduce shareholder value. More acquisition activity, other things equal, will lead to lower variance of a firm's cash flows and thus increase the value of its debt claims (Levy and Sarnat, 1970; Lewellen, 1971; Higgins and Schall, 1975). It is therefore in a creditor's interest to engage in more acquisitions, while shareholders have no specific interest in engaging in a large number of acquisitions. We measure acquisition intensity with three proxies: the probability of completing an acquisition, the number of completed acquisitions, and the value of completed acquisitions in a given year. We do not find any

<sup>1</sup> Recent literature on the composition of corporate boards points out the importance of using the correct empirical methodology in the presence of self-selection (see Adams, Hermalin, and Weisbach, 2010).

evidence for creditor presence being associated with more acquisition activity.

However, we find that those acquisitions that are completed have characteristics that are more attractive to creditors and less attractive to shareholders. Specifically, acquisitions with creditor presence tend to have deal-specific attributes that are negatively related to shareholder value: diversification into different industries (Morck, Shleifer, and Vishny, 1990), and public targets (Fuller, Netter, and Stegemolle, 2002). Furthermore, creditor presence is associated with a statistically significant 11 percent reduction in the likelihood of a cash-only payment, which may be favorable to creditors since it preserves acquirer cash holdings (Campbell, Hilscher, and Szilagyi, 2008; Acharya, Davydenko, and Strebulaev, 2011). In contrast fewer cash-only payments are unfavorable to shareholders: Cash offers tend to be characterized by insignificant abnormal stock returns while acquisitions financed by stock are associated with significantly negative returns (Travlos 1987; Wansley, Lane and Yang, 1987; Franks, Harris and Mayer, 1988; Loughran and Vijh, 1997).

Equity and credit market reactions to acquisition announcements reflect the patterns in deal attributes: We find a reduction in credit spreads, i.e. a positive credit market reaction:

Creditor presence is associated with a statistically significant 3.3 percent reduction in 5-year credit default swap (CDS) spreads when compared to firms with similar characteristics but with no creditor on their board. At the same time, creditor presence is associated with a statistically significant 4.3 percent reduction in equity value.

While these results suggest adverse wealth transfers from shareholders to creditors, one may take a stakeholder point of view and argue that so long as the total firm value that is shared between shareholders and creditors is larger as a result of an acquisition, creditors add value to the firm. Therefore, we also study the response of overall firm value to acquisitions and find that creditor presence is associated with a statistically and economically significant decrease in overall firm value.

We present further evidence in support of the conflicts of interest hypothesis. Following the empirical strategy in Kroszner and Strahan (2001) and Booth and Deli (1999), we compare creditors and unaffiliated bankers (commercial bankers with no outstanding lending relationship). While both creditors and unaffiliated bankers have financial expertise and can give sophisticated advice to management with regards to acquisition decisions, only creditors have conflicting interests with shareholders. We find no significant relationship between unaffiliated banker presence and stock, credit market, or firm value reactions to acquisition announcements.

Another group of financial experts represented on boards are investment bankers. Similar to unaffiliated bankers, investment bankers are not associated with any significant credit market response as measured by CDS spreads. This pattern is consistent with the conflicts of interest hypothesis since investment bankers (unlike creditors) have no particular interest in increasing the value of a company's bonds. On the other hand, we find that the presence of an investment-banker is associated with a 5.65 percent reduction in shareholder value around the announcement dates – a result that is consistent with previous studies that look at the relationship between investment banker presence on boards and acquisition returns (Güner, Malmendier, and Tate, 2008).

This paper adds to the literature that studies the impact of creditors on corporate boards. The relationship between creditor-directors and firm financing is the subject of a number of studies (Booth and Deli, 1999; Kroszner and Strahan, 2001; Byrd and Mizruchi, 2005; Güner et. al., 2008; Ferreira and Matos, 2011; Şişli-Ciamarra, 2011). However, to the best of our knowledge, our study is the first to look at the relationship between creditors and acquisitions — an additional avenue through which shareholder-creditor conflicts may distort corporate actions. Also, unlike previous work, we specifically analyze both equity and credit market responses and

2

<sup>&</sup>lt;sup>2</sup> Creditor representation on boards have been shown to be detrimental for shareholder value in the context of foreign bank-based financial markets: Morck and Nakamura (1999) show that banker directors emphasize policies that favor creditors over shareholders in Japan, and Dittman, Maug, and Schneider (2010) argue that bankers cause a decline in the valuations of non-financial firms in Germany.

thereby are able to get a better understanding of effects of conflicts of interest between shareholders and creditors on acquisition outcomes.

Güner, Malmendier, and Tate (2008) perform a comprehensive study of the effects of financial experts on corporate boards. As one of their outcome measures, they consider shareholder responses to acquisition announcements and find no effect of commercial banker presence. They do not, however, consider the effect of creditor presence on acquisitions nor do they analyze the credit market reaction in response to acquisition announcements.<sup>3</sup>

Our main conclusion that creditor-directors lead to value-destroying investment choices (acquisitions) is consistent with Mitchell and Walker (2008), who show that firms appointing creditors on their boards exhibit value declines following their appointments. Our results add to Kracaw and Zenner (1999), who show that the stock price reactions to bank loan announcements are negative when the bank and the borrower have representation on each others' boards.

The rest of the paper is organized as follows. Section 2 describes our data and section 3 motivates and discusses our empirical methodology. Section 4 presents our results of the effect of creditor presence on acquisitions. Section 5 presents further evidence in support of the conflicts of interest hypothesis. Section 6 concludes.

### 2. Data description and summary statistics

#### 2.1. Sample and board of directors data

Our sample consists of all companies included in the S&P 500 Index at the beginning of 2002, excluding financial companies (SIC codes 6000-6999). We gather data on these companies for six years between 2002 and 2007. Conditional on available information our sample includes

<sup>&</sup>lt;sup>3</sup> Güner et al. (2008) consider differences between creditors and unaffiliated bankers for other outcome measures, e.g. loan sizes, though not when studying acquisitions.

398 companies and 2230 firm-year observations. For the companies in our sample we hand-collect information on individual director characteristics from company annual reports and proxy statements filed with the Securities and Exchange Commission (SEC). Annual reports list the members of the board of directors, and proxy statements contain their biographies, from which we identify whether the primary employer of a director is a commercial bank (banker-director henceforth). We classify a financial institution as a commercial bank if the Federal Deposit Insurance Company (FDIC) lists it as a U.S. chartered commercial bank. Proxy statements also contain detailed information on the employment histories of directors so that we can identify the years for which the director is employed at the bank. We also collect other information about the structure of boards such as the number of board members, the number of insiders, and the tenures of the board members.

#### 2.2. Classification of banker-directors on boards

Having identified the members of the board that are bankers, we classify them as either creditors or unaffiliated banker-directors using data on loan contracts from the Reuters/Loan Pricing Corporation's (LPC) Dealscan database. The data includes the names of all the banks in the lending syndicate at the time of origination; the loan contract date; the amount, maturity, type, and purpose of the loan; and information on the price and non-price terms of the loan contract.

A board member is classified as a creditor if she is an executive of a commercial bank that has extended at least one loan to the company over the previous five years as a sole lender, or as a lead arranger in a syndicate. We require the creditor-director to be employed at one of the

<sup>4</sup> The sample does not include observations for each firm-year because we condition on the availability of explanatory variables and because some firms in the 2002 S&P 500 drop out of our sample due to mergers, acquisitions, and going private. The patterns in banker presence over time are the same when including all 403 non-financial companies for which we have collected data on board composition. The list of firms included in our analysis is available from the authors.

<sup>&</sup>lt;sup>5</sup> Public companies are required to disclose the employment histories of their board members and nominees for the previous five years. In general, most companies release the employment histories of their directors for a much longer time, starting from their initial employments.

<sup>&</sup>lt;sup>6</sup> Dealscan is the most comprehensive and up-to-date source for stand-alone and syndicated loans. See Carey, Post, and Sharpe (1998) for a more detailed description of Dealscan.

lead banks in the syndicate, because the ex-ante due diligence and the ex-post monitoring of a loan is delegated to the lead bank within a syndicate. If there is no lending relationship over the previous five years between a company and a bank where the director is employed, we classify the banker as an unaffiliated banker.

Table 1 summarizes the percentage of companies with at least one commercial banker serving on their board of directors. For each year in our sample (2002 to 2007) we report the percentage of firms with a commercial banker on the board. We see that despite the separation of banking and commerce in the United States, commercial banks are actively engaged in the corporate governance of non-financial companies by holding directorships: bankers are present on boards for 28.9 percent of the sample (company-years), and creditors are present for 11.4 percent of observations. We also report the share of firms with an investment banker present on the board; it is equal to 10 percent and stable over time.

From 2004 to 2005 there is a drop in commercial banker presence on boards, which may be attributed to changes in the regulatory environment shaping the composition of boards of directors following the passage of the Sarbanes-Oxley Act in July 2002. By the end of 2004 large public corporations were scheduled to comply with the new regulations, which dictated the elimination of inside directors from certain board committees and strongly discouraged the presence of affiliated directors. For example, Citigroup executive Robert Rubin resigned from the board of directors of Ford, stating that "Citigroup's multifaceted relations with Ford could raise a question whether my relationship with Ford and Citigroup creates an appearance of conflict." However, subsequently (from 2005 to 2007) there is a rebound in the representation of

7

<sup>&</sup>lt;sup>7</sup> Our findings are consistent with the previous literature. Booth and Deli (1999) find 22.2% of non-financial S&P 500 constituent firms to have a commercial banker serving on their boards in 1990. In 1992, 31.6% of the Forbes 500 companies include a director who is a commercial banker, and 5.8% have a director who is an executive of their main lender (Kroszner and Strahan, 2001). For 2000, the percentage of S&P firms with a banker on board is 25 percent (Santos and Rumble, 2004). Güner, Malmendier, and Tate (2008) find that 21 percent of firms in their sample (1988-2001) had unaffiliated bankers on their boards and between 1988 and 2001, 6 percent had creditors on the board. Thus creditor presence on boards has become more prominent over time.

<sup>&</sup>lt;sup>8</sup> See AP 08/25/2006.

commercial bankers on company boards. The ongoing high level of creditor presence highlights the importance of analyzing whether or not potential conflicts of interest between creditors and shareholders result in value-destroying decisions or if instead creditors provide services on corporate boards that are valuable to the shareholders.

#### 2.3. Acquisition data

In order to analyze acquisition level data, such as deal attributes and market reactions to announcements, we gather information for all completed acquisitions undertaken by the companies in our sample. We include acquisitions with announcement dates and effective dates between January 1, 2002, and December 31, 2007 in the Mergers and Acquisitions database of ThomsonOne Banker. We include both public and private targets and, as is standard, exclude repurchases. Conditional on available equity and credit market data as well as firm- and deal-specific controls, the acquisition sample contains 1641 acquisitions. Table 2 lists the number of acquisitions with a creditor, an unaffiliated banker, or an investment banker present on the board of directors of the acquirer. For 15.7 percent of acquisitions there was a creditor, for 14 percent there was an unaffiliated banker, and for 10.1 percent there was an investment banker present on the board of the acquirer.

To this data we add measures of firm characteristics and market reactions to acquisition announcements. We use accounting data from COMPUSTAT, equity market data from CRSP, and CDS spread data from Markit. Variable definitions and data sources are summarized in the appendix.

## 3. Empirical methodology: Correcting for selection

We consider three main questions in this paper: (i) Is creditor presence associated with a higher propensity to acquire? (ii) Do companies with creditors on their boards acquire targets

that have attributes that benefit creditors? (iii) What are the wealth effects of acquisitions on the shareholders and creditors when there is a creditor on an acquirer's board? In each case the main equation of interest takes the form

$$Y_{i} = \beta_{0} + \beta_{1}X_{i} + \beta_{2}D_{i} + e_{i}, \tag{1}$$

where  $Y_i$  measures acquisition intensity, acquisition deal attributes, or cumulative abnormal returns for shareholders and creditors of firm i.  $X_i$  is the set of control variables, and  $D_i$  is a dummy variable indicating the presence of a creditor on the company's board.

In estimating these regressions we explicitly take into account the possibility of bankers self-selecting onto boards of directors, because previous research on banker-directors suggests that the set of companies with a banker on their board are not a random sample. Booth and Deli (1999), Kroszner and Strahan (2001), Byrd and Mizruchi (2005) and Şişli-Ciamarra (2011) find that the probability of having a banker on the board is a function of the firm's size, the extent of information asymmetry (proxied for by market-to-book ratio, research and development expenditure, and presence of short-term credit rating), and previous debt levels. In addition, self selection has become an important empirical concern in studies of corporate boards (Li and Prabhala, 2007; Adams, Hermalin, and Weisbach, 2010).

Consistent with previous work, we find that there are important differences in financial characteristics between companies with and without creditor board presence. Table 3 reports averages of firm-level characteristics for the sample of firm-year observations. We report averages for the full sample and for sub-samples of company-years with and without creditor presence and also test if differences in means are statistically significant.

Creditors serve on the boards of companies that are larger, have higher leverage, lower return volatilities, lower R&D expenditures, lower market-to-book ratios, and a higher likelihood of having a short-term credit rating. Boards with creditors are significantly larger and more independent (fewer insiders). Furthermore, creditors are not serving on the boards of companies

with higher levels of financial distress, as measured by 5-year CDS spread, S&P credit rating, and annual default probability (Campbell, Hilscher, and Szilagyi, 2008). Instead, lower CDS spreads and lower-risk credit ratings suggest that creditor presence is associated with lower credit risk. 

These results are in agreement with the previous literature including Kroszner and Strahan (2001), who argue that creditors tend to join the boards of financially more stable companies.

We also compare differences in firm financial characteristics and deal-specific variables for the sample of acquisitions. Table 4 reports average characteristics for the full sample and for acquisitions with and without creditor-director presence. Acquirers with creditors have relatively larger cash holdings, lower market-to-book ratios, and lower stock return volatilities. Companies with creditor-directors also have lower levels of credit risk, though differences are not statistically significant. There are also differences in deal attributes: firms with creditors engage in more diversifying and foreign target acquisitions, and significantly fewer cash-only deals. We study these deal attributes in detail when performing multivariate analysis in Section 4.

Differences between firms with and without creditors imply that creditors may be self-selecting onto corporate boards and that OLS estimates for equation (1) may be inconsistent. If there is selection and  $D_i$  is correlated with the error term  $e_i$ , this results in a biased estimator for  $\beta_2$  and also for the other coefficients (Greene, 2003). To correct for this we use the average treatment effects model that is developed specifically to analyze the case of an endogenous dummy variable (Heckman, 1978). The probability of receiving the treatment is modeled together with the structural outcome equation. The full model (Greene, 2003, pp. 787-788) is:

$$Y_i = \beta_0 + \beta_1 X_i + \beta_2 D_i + e_i$$
 (2a)

$$D_i^* = \delta Z_i + u_i \tag{2b}$$

$$D_i = 1 \text{ if } D_i^* > 0$$

<sup>&</sup>lt;sup>9</sup> In theory it might be possible that if a firm becomes financially distressed, a bank with an outstanding lending relationship asks one of its employees to join the board of the firm. We do not see any evidence consistent with this hypothesis. Instead, bankers often join boards before they become creditors.

$$D_i = 0 \text{ if } D_i^* < 0$$

where  $D_i$  is an endogenous dummy variable indicating whether or not the treatment is received (whether or not a creditor is present on the board). The binary decision to obtain the treatment is modeled as an outcome of an unobserved latent variable,  $D_i^*$ .  $Z_i$  is a set of characteristics that affect the likelihood of the treatment – in this study whether or not a creditor is present on the board of directors. The individual error terms,  $e_i$  and  $u_i$ , are assumed to have a bivariate normal distribution:

$$\varepsilon_i \sim N(0,\sigma)$$

$$u_i^N(0,1)$$

$$corr(e_i,u_i)=\rho$$
.

In specifying the treatment equation (2b), we rely on the findings of Booth and Deli (1999), Kroszner and Strahan (2001), Byrd and Mizruchi (2005), Rumble and Santos (2006), and Şişli-Ciamarra (2011), who examine the determinants of banker presence on a company's board of directors. The variables that explain banker-director presence include firm size, total debt, short-term debt, tangible assets, cash holdings (all scaled by assets), as well as stock return volatility, market-to-book ratio, research and development expenditure, default probability, debt rating, ratio of insiders, board size, year dummies, and industry dummies. Board composition does not adjust immediately to changes in a firm's environment but rather with a lag, which is true especially if the board of directors is structured as staggered. We therefore average all explanatory variables in the treatment equation over the past three years.

We discuss the variables included in the structural equation (2a) in Section 4 when we discuss the effect of creditor presence on acquisition activity, types of acquisitions, as well as credit and equity market reactions. We note that in the case of non-linear treatment models the set of explanatory variables included in the structural equation (2a) need not be a subset of the

<sup>10</sup> A staggered board of directors occurs when a corporation elects only a portion of its directors at a time, with different groups of directors having overlapping multi-year terms, usually three-year terms.

variables in the treatment equation (2b). We therefore in all cases choose appropriate explanatory variables based on the relevant literatures (e.g. determinants of acquisition activity or market reactions). What this also means is that we need not be concerned with exclusion restrictions which, though strictly speaking not necessary in non-linear treatment models (Heckman and Navarro-Lozano, 2004), may be useful in practice (Li and Prabhala, 2007). For example, the treatment equation contains board size, which is associated with banker presence (see Table 3), but the structural equation (2a) does not. Furthermore, since all explanatory variables in the treatment equation are averaged over the past three years, they are different from the explanatory variables in the structural equation.

#### 3.1. Estimation strategy

Statistically significant differences in financial characteristics of firms with and without creditors on their boards suggest that self-selection may be a concern. Therefore, for each model specification we test whether or not there is evidence of self-selection. In average treatment effects estimation, a statistically significant correlation between the error terms of the structural and treatment equations  $(\rho)$  represents statistical evidence for the presence of self-selection. For each regression specification we therefore test whether or not the correlation coefficient between the error terms is significantly different from zero. If there is evidence of self-selection, specifically, if the correlation coefficient is significant at the 5% level, we estimate the average treatment effects model that controls for it. If there is no such evidence, we use the standard OLS model. Several of the variables we are interested in are binary (e.g. the probability of a company making a diversifying acquisition). For these variables we estimate bivariate probit models instead of using an average treatment effects model.

#### 4. Creditor-directors on boards and acquisitions

Creditors benefit from corporate actions that decrease the probability of default. One such action is the acquisition of another company with imperfectly correlated earnings and valuation. Such an acquisition will likely result in a combined firm with lower volatility (coinsurance effect) and therefore increase the value of the company's debt. However, an acquisition may add value to creditors while not increasing shareholder value and therefore potentially lead to conflicts of interest. Thus, if conflicts of interest between shareholders and creditors manifest themselves in acquisition decisions when a creditor serves on a firm's board, we expect such firms to either acquire more or to make acquisitions that benefit creditors but not necessarily shareholders. We first investigate both of these possibilities. We then consider equity and credit market reactions directly.

#### 4.1. Acquisition activity

We measure acquisition activity using three dependent variables: (i) *Acquisition dummy* equals one if a firm undertakes at least one acquisition in a given fiscal year, (ii) *Acquisition count* is the number of acquisitions completed in a given fiscal year, (iii) *Acquisition value / total assets*, the relative value of acquisitions, is the ratio of total acquisition value to lagged firm size in a given fiscal year.

Table 5 Panel A reports means for the three measures along with comparison of means tests. During our sample period, in a given year an average company has an unconditional probability of 47.5 percent of completing an acquisition and the average size of acquisitions is 2.9 percent of total assets. Differences across groups with and without a creditor on the board are not statistically significant. The average number of acquisitions per fiscal year is 1.3 (creditor present) compared to 1.1 (no creditor) and this difference is significant at the 10 percent level.

<sup>11</sup> In the seminal Merton (1974) model of corporate debt the credit spread is an increasing function of asset volatility.

<sup>&</sup>lt;sup>12</sup> A merger may also positively affect the asset value of either of the companies, for example due to increased synergies. We consider shareholder, creditor and overall firm value responses to acquisition announcements in Section 4.3.3.

The slightly higher number of acquisitions by companies with creditors is consistent with the higher share of acquisitions with creditor presence (15.7%), when compared to the overall fraction of 11.4% of companies with creditors (Table 1). As discussed, these differences may be the direct result of creditor presence but they may also be due to the nature of creditor selection onto corporate boards. For example, larger firms tend to acquire more and are also more likely to have a creditor on their board.

Table 5 Panel B presents results from multivariate regressions, which control for other possible determinants of acquisition activity: Size, market-to-book ratio, cash to assets, capital expenditures to assets, leverage ratio, as well as the governance index, <sup>13</sup> a proxy of the overall corporate governance quality of the acquirer, and ratio of insiders. All regressions include year and industry effects (1 digit SIC codes), and standard errors are clustered at the firm level. The regressions test and control for self-selection if it is present.

We find that creditor presence does not affect acquisition probability, acquisition count or acquisition relative value (columns 1 to 3). The results imply that creditor presence is not associated with increased acquisition intensity.

#### 4.2. Types of acquisitions

What might matter more than acquisition quantity is acquisition quality: Creditors may try to gain by favoring acquisitions with deal attributes that are associated with higher creditor value, and this may come at the expense of shareholder value. Therefore, we next analyze the effects of creditor presence on certain acquisition deal attributes that may be more favorable to creditors: (i) targets in unrelated industries, (ii) foreign targets, (iii) public targets, and (iv) cash-financed acquisitions. We expect creditors to favor a higher likelihood of the first three and a lower frequency of cash deals. Table 6 reports results using a multivariate framework. The first

<sup>13</sup> See Gompers, Ishii, and Metrick (2003). The Governance Index is a summary measure of corporate governance, based on 24 governance factors provided by the Investor Responsibility Research Center.

set of control variables are acquirer's financial characteristics and include size, market-to-book ratio, cash-to-assets ratio, leverage, prior stock performance, and stock return volatility during the 12 months preceding the acquisition announcement. The second set of control variables relates to deal-specific characteristics: if the target is public or foreign, whether or not the acquisition is diversifying, if the bid is hostile, and what the method of payment is. The final set of control variables includes the ratio of insiders, and the governance index.

#### 4.2.1. Diversifying acquisitions: targets in unrelated industries

Creditor presence leads to an increase in diversifying acquisitions, which we define as acquisitions of targets in unrelated industries (two-digit SIC code of the acquirer is different from that of the target). Creditors may benefit from diversifying acquisitions because they will likely lead to a reduction in the overall risk of the company. In contrast, diversifying deals are often viewed as being value-destroying for shareholders (e.g., Morck, Shleifer, and Vishny, 1990).

In our sample 47 percent of acquisitions are diversifying when there is no creditor (Table 4) while 61 percent of acquisitions with a creditor involve targets in an unrelated industry.

Controlling for firm, deal, and governance measures, creditor-director presence is associated with a 10.4 percent<sup>14</sup> increase in the probability of a diversifying acquisition, which is significant at the 10 percent level (Table 6, column 1).

#### 4.2.2. Foreign targets

Acquirers may achieve diversification not only by acquiring companies in unrelated industries, but also by acquiring foreign (non-U.S.) companies and achieving international diversification in their operations. Table 4 shows that creditor presence is associated with a higher frequency of foreign target acquisition (36 percent) compared to no creditor (31 percent); the difference in means is statistically significant at the 10% level. In the multivariate setting

<sup>&</sup>lt;sup>14</sup> In the table we report probit coefficients. For ease of interpretation we discuss marginal effects in the text.

creditor presence is associated with a 3.8 percent increase in the probability of acquiring a foreign target, though the coefficient is not statistically significant (Table 6, column 2).

#### 4.2.3. Public targets

Creditor presence is associated with a 3.4 percent increase in the tendency to acquire a public target. The coefficient is significant at the 10 percent level (Table 6, column 3). The higher propensity to acquire public targets is important for shareholders since acquisition of a public target is associated with lower returns to the acquirers' shareholders (Fuller, Netter, Stegemolle, 2002).

#### 4.2.4. Cash-only payment

The final deal attribute we analyze is the acquisition payment terms. Cash offers are characterized by insignificant abnormal stock returns while acquisitions financed by stock have significantly negative returns (see Travlos, 1987; Wansley, Lane, and Yang, 1987; Franks, Harris, and Mayer, 1988; Loughran and Vijh, 1997; as well as Heron and Lie, 2002 for a review of the literature). Table 4 shows that acquisitions paid for entirely in cash are less frequent in the presence of a creditor (18 percent) compared to no creditor on the board (26 percent). This effect persists in the multivariate results (Table 6, column 4). Creditor presence is associated with a 10.1 percent reduction in the probability that the deal is financed entirely by cash, and this result is significant at the 0.1 percent level.

We note that lower levels of cash financing rules out the hypothesis that companies with creditors on their boards are more likely to acquire since creditors provide loans, perhaps as a favor in return for the board seat. Such increased availability of cash might indirectly help the acquirer finance unnecessary and excessive cash-financed acquisitions. In addition, the results in Table 5 show that there is no evidence for creditor presence leading to higher acquisition intensity. We have also checked that there are no instances of an acquirer obtaining a loan from the director's bank in order to finance the acquisitions (bridge financing).

To summarize, we find that creditor-director presence is associated with a higher likelihood of deal attributes that are less favorable for acquirer shareholders and more favorable for acquirer creditors: more diversification, more public targets, and less cash-only financing.

#### 4.3. Market reactions to acquisition announcements

We next consider market reactions to acquisition announcements and show that: (i) the equity market reaction to acquisition announcements is more negative if a creditor is present, consistent with the higher likelihood of acquisitions that have been shown to be value-destroying for shareholders; (ii) the credit market reacts more favorably to acquisitions announcements with creditor-director presence, consistent with higher levels of diversifying and cash-preserving acquisitions that we expect to be value-enhancing for creditors; and (iii) the overall firm value is negatively affected by creditor presence. The results are consistent with the conflicts of interest hypothesis.

#### 4.3.1. Equity market reaction: Negative

We examine market reactions using an event study. We calculate cumulative abnormal returns (CARs) during the eleven-day window (t-5, t+5) around acquisition announcement dates:

$$CAR_{i,t-5,t+5} = \prod_{k=t-5}^{t+5} (1 + R_{ik} - R_{mk}) - 1,$$
 (3)

where  $R_{it}$  is the return on the common stock of firm i on day t, and  $R_{mt}$  is the return of the CRSP value-weighted market index for day t. We then compare returns for acquisitions with and without a creditor-director.

Table 7 Panel A reports average cumulative abnormal returns for the full sample and for the groups with and without a creditor on the board. The average equity return is -0.13 percent.

The slightly negative and insignificant average return for acquirers is in line with the literature on

17

<sup>&</sup>lt;sup>15</sup> We calculate excess returns in this way to reduce noise from estimating market betas. We have checked that the results are robust to calculating abnormal returns using the single factor model (market return), where we estimate betas for the 120 trading days ending 10 days prior to the announcement.

mergers and acquisitions (see Andrade, Mitchell, and Stafford, 2001; Bruner, 2002; Tuch and Sullivan, 2007). When a creditor is on the board, the return is slightly larger (0.39 percent) than when there is no creditor (-0.23 percent), though the difference is not statistically significant.

Table 7 Panel B (column 1) presents results from a multivariate analysis, controlling for variables that may influence the equity market reaction to acquisition announcements, and correcting for selection, which is statistically significant at the 1 percent level. Creditor presence is associated with a statistically and economically significant decrease in shareholder wealth: the cumulative abnormal return is 4.3 percent lower when there is a creditor on the board, and the effect is significant at the 5% level.

Since one of our contributions is to highlight the importance of correcting for selection, we briefly discuss the difference between the uni- and multivariate analysis (Panel A and Panel B): The test of independent equations ( $\rho$ ) is rejected at the 1 percent level, which means that there is statistically significant evidence of self-selection. The correlation coefficient is positive (0.53), which implies that the OLS estimate of the coefficient on the creditor dummy (not controlling for self-selection) is biased upward. If we do not control for selection, we are less likely to find a negative relationship between equity returns and creditor presence, consistent with the economically and statistically insignificant unconditional difference in means.

#### 4.3.2. Credit market reaction: Positive

We next examine the credit market reaction to acquisition announcements. Following Wei and Yermack (2010) we use percent changes in 5-year CDS spreads to measure the credit market reaction. We calculate spread changes over the eleven day period (t-5, t+5) around the announcement date; similar to the equity return adjustment we subtract the contemporaneous average spread change in order to adjust for overall credit market movements.

1

<sup>&</sup>lt;sup>16</sup> Wei and Yermack (2010) use CDS spread percent changes (from Markit) to analyze the credit market reaction to the release of CEO inside debt holdings in proxy statements. Acharya and Johnson (2007) and Hilscher, Pollet, and Wilson (2011) also use CDS percent changes to measure information in the credit market.

Table 7 Panel A reports unconditional average spread changes. For the full sample there is a slight increase in CDS spreads (0.54 percent, marginally statistically significant at the 10 percent level), consistent with Billett, King, and Mauer (2002) who find that the average corporate bond reaction of the acquirer is negative. CDS spreads increase by more if there is no creditor (0.59 percent), compared to virtually unchanged spreads if there is a creditor (0.2 percent, not significant). In Panel B (column 2) we perform a multivariate analysis of the credit market reaction and find that creditor presence is associated with a statistically significant decrease in CDS spreads of 3.3 percent (good news for creditors).<sup>17</sup>

This evidence supports the hypothesis that creditors on boards may prompt management to engage in acquisitions that increase the value of creditors' employers' loan book (a reduction in credit spreads). Creditors can gain either from acquisitions that increase overall firm value or from those that do not increase firm value but reallocate wealth from shareholders to bondholders. The negative and significant effect of creditor presence on shareholder value (column 1) together with the positive and significant effect on creditor value (column 2) is consistent with such wealth transfers.

#### 4.3.3. Total firm value reaction: Negative

The results for shareholder and creditor reactions suggest adverse wealth transfers. It is also useful to consider a stakeholder point of view and ask how total firm value is affected by creditor presence. Therefore, we next measure the response of overall firm value by calculating the weighted average of equity and debt returns. <sup>18</sup> The equity return is the cumulative abnormal return in (3); we measure the bond return as the return on a 5-year bond that has a credit spread over U.S. Treasuries equal to the CDS spread. To adjust for debt market movements we

<sup>&</sup>lt;sup>17</sup> This result is in line with the results of Renneboog and Szilagyi (2009), who show that in a more creditor-friendly jurisdiction, bond performance around M&A announcements is higher.

<sup>&</sup>lt;sup>18</sup> Firm value is equal to the market value of equity plus debt. We therefore need to calculate the firm's debt returns, which are not equal to the spread change used in the previous section.

subtract the average return to calculate abnormal bond returns. <sup>19</sup> We then use leverage, measured as the book value of long term debt divided by the sum of long term debt and market equity, and one minus leverage as weights on debt and equity returns:

$$Firm\_ret_{t-5,t+5} = (1 - leverage) * CAR_{equity,t-5,t+5} + leverage * CAR_{debt,t-5,t+5}.$$

$$(5)$$

Table 7 Panel A reports unconditional means of firm returns. Consistent with the patterns in equity returns, the average firm return is slightly higher when there is a creditor on the board (0.28 percent) than when there is not (-0.26 percent). However, once we control for selection and other explanatory variables (Table 7 Panel B), we find a negative effect of creditor presence on the firm return of -5.1 percent, which is statistically significant at the 0.1% level.

To summarize, we find evidence in support of the conflicts of interest hypothesis. The presence of creditors on boards result in acquisitions that decrease shareholder value, increase creditor value, and lead to a decrease in overall firm value. The evidence is consistent with the increased tendency of firms with creditor presence to engage in deals with attributes that we expect to be more favorable for creditors, such as more diversifying acquisitions and acquisitions that are less likely financed with cash.

Furthermore, the evidence is inconsistent with the hypothesis of enhanced monitoring, the idea that creditor-directors may be effective monitors of management and that they add value to the board decisions due to their lending relationship with the company and resulting informational advantage. Creditors may thus be both interested and capable to screen out bad investment proposals and select value-enhancing investment projects (Leland and Pyle, 1974; Campbell and Kracaw, 1980; Diamond, 1984; Boyd and Prescott, 1986). The less favorable outcomes associated with creditors do not support the enhanced monitoring hypothesis.

\_

<sup>&</sup>lt;sup>19</sup> We use the change in price of a constant maturity 5-year bond trading at par to proxy for the bond return. The yield to maturity is set equal to the 5-year constant maturity U.S. Treasury bond plus the 5-year CDS spread.

We note that our findings cannot be explained by creditors' banks providing financing or acting in advisory capacities. First, creditor-directors do not provide financing for acquisitions that are in our sample. Second, there are no instances where the creditor's bank acts as an advisor to the acquirer or to the target. Moreover, in rare cases more than one creditor serves on an acquirer's board of directors. Our results are robust to excluding these observations.

# 5. Further evidence for conflicts of interest: The role of non-conflicted financial experts

#### 5.1. Unaffiliated banker-directors

Having documented that creditor presence is associated with less favorable acquisition outcomes for shareholders, we now present further supporting evidence for the conflicts of interest hypothesis. We follow the empirical strategy suggested by Kroszner and Strahan (2001) and Booth and Deli (2003) and turn our attention to unaffiliated banker directors. Unaffiliated bankers possess financial expertise and may enable the company to pick acquisitions that create value. As experts in financial matters, they can use their knowledge to calculate the reservation prices for target firms, evaluate the potential gains from synergies between operations of acquirers and targets, and analyze the risks associated with merger transactions. However, compared to the creditors, they do not have conflicts of interests with shareholders.

We consider the effect of unaffiliated banker presence on equity and credit market responses to acquisitions. 14 percent of the acquisitions in our sample involve unaffiliated banker-directors (Table 2). Since acquisitions with unaffiliated banker presence tend to be undertaken by companies that are more stable (Table 8), we continue to implement tests for evidence of selection and correct for selection if it is present.

Table 9 presents results from a multivariate analysis of market reactions to acquisition announcements. In contrast to creditors, unaffiliated banker presence does not lead to less favorable stock market reactions (column 1). Returns to shareholders are statistically insignificant and close to zero. Unaffiliated banker presence also does not affect credit market outcomes or result in lower levels of firm value (columns 2 and 3). In both cases coefficients are close to zero and not statistically significant. Unaffiliated banker presence on boards is thus associated with acquisition outcomes that do not decrease shareholder value when compared to creditor-director presence.<sup>20</sup> This evidence is consistent with the conflicts of interest hypothesis.

#### 5.2. Investment bankers

We also consider the impact of investment bankers on boards and show that their presence is associated with less favorable equity market and overall firm reactions to acquisitions but do not benefit creditors. Investment bankers in our sample are executives of pure investment banks and do not include bank holding companies that provide commercial banking and investment banking services at the same time.

Table 10 reports results from analyzing the effect of investment banker presence on market reactions to acquisition announcements. We find a statistically significant relationship of -5.7 percent between investment banker-director presence and stock returns around the announcement days. <sup>21</sup> Our results are consistent with Güner et al. (2008) who study investment bankers as conflicted financial experts. Our results add to their findings since we control for selection (statistically significant at the 0.1% level) and directly examine credit and firm value responses. For equity returns, the direction of self-selection ( $\rho$ >0), indicates that the OLS coefficient is biased upward: controlling for selection the effect of investment banker presence is

٠

<sup>&</sup>lt;sup>20</sup> Consistent with the market reaction results, unaffiliated banker presence is also not associated with patterns in acquisition intensity and deal attributes that favor either creditors or shareholders. The (unreported) results are available from the authors.
<sup>21</sup> The negative equity response is consistent with the (unreported) effect of investment banker presence on acquisition behavior and deal attributes: investment banker presence is associated with significantly larger acquisitions and significantly more acquisitions of foreign targets. Results are available from the authors.

therefore more negative as compared to Güner et al. Investment banker presence is not associated with any significant response in the credit market (column 2), similar to unaffiliated banker-director presence. However investment banker presence is associated with a 3.7 percent reduction in total firm value (column 3).

Our results support the conflicts of interest hypothesis and add to our understanding of investment banker presence on boards. Unlike investment bankers, creditors, presumably motivated to preserve the value of their employers' loans to the company, affect credit market returns positively. However, both groups are associated with a reduction in shareholder value.

In an attempt to identify possible direct conflicts of interest with shareholders, we check if the investment banks that employ the directors in our sample provide any advisory services to the acquirers or to the targets in our sample, and do not find any such relationships. Thus the evidence suggests that, even though there are no direct conflicts of interest, investment bankers presence results in acquisitions associated with poor equity and firm-value outcomes.

#### 6. Conclusion

Creditor representation on corporate boards is substantial. Yet, little is known about how it affects corporate behavior. In this study we present evidence consistent with the hypothesis that conflicts of interest lead to wealth transfers from shareholders to creditors (Galai and Masulis, 1976). We analyze acquisition activity, deal attributes and explicitly consider equity, credit, and overall firm value reactions to acquisition announcements using a hand-collected data set for S&P500 non-financial companies from 2002 to 2007. We present evidence of creditor selection onto corporate boards: Boards with and without creditor-directors are different in ways that can bias estimation results. We therefore correct for selection following Heckman (1978).

Creditor presence on a company's board of directors is associated with less favorable

equity market reactions, more favorable credit market reactions, and a reduction in overall firm value. Companies with creditor-directors tend to engage in acquisitions that are more diversifying, acquiring more targets in unrelated industries and public targets, as well as a substantially lower probability of financing the acquisition with cash. The unfavorable equity market reactions are consistent with the deal attributes, which the previous literature has shown to be associated with negative equity returns. Increased diversification and preservation of cash are consistent with the favorable credit market outcomes. The evidence is thus consistent with the conflicts of interest hypothesis.

We present additional evidence in support of the conflicts of interest hypothesis:

Unaffiliated bankers, commercial bankers whose employers do not have an ongoing lending relationship with the company, are not associated with negative shareholder or firm-value reactions, nor does their presence result in positive credit market responses. Finally, we show that the presence of investment banker-directors, potentially conflicted financial experts, are associated with negative equity returns but not, in contrast to creditors, favorable credit market outcomes.

Our results add to the ongoing literature on mechanisms of corporate governance.

Conflicts of interest between shareholders and creditors are an important potential source for sub-optimal decision making. We show that in the case of acquisition decisions board composition should be a concern.

#### References

Acharya, Viral V., Sergei A. Davydenko, and Ilya A. Strebulaev, 2011, Cash holdings and credit risk, NBER working paper 16995.

Acharya, Viral V. and Timothy C. Johnson, 2007, Insider trading in credit derivatives, *Journal of Financial Economics* 84, 110-141.

Adams, Renee B., Benjamin E. Hermalin, and Michael S. Weisbach, 2010, The role of boards of directors in corporate governance: A conceptual framework and survey, *Journal of Economic Literature* 48, 58-107.

Andrade, Gregor, Mark Mitchell, and Erik Stafford, 2001, New evidence and perspectives on mergers, *Journal of Economic Perspectives* 15, 103-120.

Billett, Matthew T., Tao-Hsien Dolly King, and David C. Mauer, 2004, Bondholder wealth effects in mergers and acquisitions: New evidence from the 1980s and 1990s, *Journal of Finance* 59, 107-135.

Booth, James R., and Daniel N. Deli, 1999, On executives of financial institutions as outside directors, *Journal of Corporate Finance* 5, 227-250.

Boyd, John H., and Edward C. Prescott, 1986, Financial intermediary-coalitions, *Journal of Economic Theory* 38, 211-232.

Bruner, Robert F., 2002, Does M&A pay? A survey of evidence for the decision-maker, *Journal of Applied Finance* 12, 48-68.

Byrd, Daniel T., and Mark S. Mizruchi, 2005, Bankers on the board and the debt ratio of firms, *Journal of Corporate Finance* 11, 129-173.

Campbell, John Y., Jens Hilscher, and Jan Szilagyi, 2008, In search of distress risk, *Journal of Finance* 63, 2899-2939.

Campbell, Tim S., and William A. Kracaw, 1980, Information production, market signalling, and the theory of financial intermediation, *Journal of Finance* 35, 863-882.

Carey, Mark, Mitch Post, and Steven A. Sharpe, 1998, Does corporate lending by banks and finance companies differ? Evidence on specialization in private debt contracting, *Journal of Finance* 53, 845-878.

Diamond, Douglas W., 1984, Financial intermediation and delegated monitoring, *Review of Economic Studies* 51, 393-414.

Dittmann, I., E. Maug, and C. Schneider, 2010, Bankers on the boards of German firms: What they do, what they are worth, and why they are (still) there, *Review of Finance* 14, 35-71.

Ferreira, Miguel A. and Matos, Pedro P., 2011, Universal Banks and Corporate Control: Evidence from the Global Syndicated Loan Market, Available at SSRN: http://ssrn.com/abstract=1364190

Franks, Julian R., Robert S. Harris, and Colin Mayer, 1988, Means of payment in takeovers:

Results for the United Kingdom and the united states, in Alan J. Auerbach, ed.: *Corporate takeovers: Causes and consequences* (National Bureau of Economic Research Project Report series Chicago and London: University of Chicago Press).

Fuller, Kathleen, Jeffry Netter, and Mike Stegemoller, 2002, What do returns to acquiring firms tell us? Evidence from firms that make many acquisitions, *Journal of Finance* 57, 1763-1793.

Galai, Dan, and Ronald W. Masulis, 1976, The option pricing model and the risk factor of stock, *Journal of Financial Economics* 3, 53-81.

Greene, William H., 2003, Econometric Analysis (Upper Saddle River, N.J.: Prentice Hall).

Gompers, Paul, Joy Ishii, and Andrew Metrick, 2003, Corporate governance and equity prices, *Quarterly Journal of Economics* 118, 107-155.

Güner, A. Burak, Ulrike Malmendier, and Geoffrey Tate, 2008, Financial expertise of directors, *Journal of Financial Economics* 88, 323-354.

Heckman, James J., 1978, Dummy Endogenous Variables in a Simultaneous Equation System, *Econometrica* 46, 931-959.

Heckman, James, and Salvador Navarro-Lozano, 2004, Using matching, instrumental variables, and control functions to estimate economic choice models, *Review of Economics and Statistics* 86, 30-57.

Heron, Randall, and Erik Lie, 2002, Operating performance and the method of payment in takeovers, *Journal of Financial and Quantitative Analysis* 37, 137-155.

Higgins, Robert C., and Lawrence D. Schall, 1975, Corporate bankruptcy and conglomerate merger, *Journal of Finance* 30, 93-113.

Hilscher, Jens, Joshua M. Pollet and Mungo Wilson, 2011, Are credit default swaps a sideshow? Evidence that information flows from equity to CDS markets, Working Paper.

Jensen, Michael C., and William H. Meckling, 1976, Theory of the firm: Managerial behavior, agency costs and ownership structure, *Journal of Financial Economics* 3, 305-360.

Kracaw, William, and Marc Zenner, 1999, Bankers in the Boardroom: Good News or Bad News?, Working Paper.

Kroszner, Randall S., and Philip E. Strahan, 2001, Bankers on boards: Monitoring, conflicts of interest, and lender liability, *Journal of Financial Economics* 62, 415-452.

Leland, Hayne E., and David H. Pyle, 1977, Informational asymmetries, financial structure, and financial intermediation, *Journal of Finance* 32, 371-387.

Levy, Haim, and Marshall Sarnat, 1970, Diversification, portfolio analysis and the uneasy case for conglomerate mergers, *Journal of Finance* 25, 795-802.

Lewellen, Wilbur G., 1971, A pure financial rationale for the conglomerate merger, *Journal of Finance* 26, 521-537.

Li, Kai, and Nagpurnanand Prabhala, 2007, Self-selection models in corporate finance, in Espen, Eckbo, ed.: Handbook of Corporate Finance: Empirical Corporate Finance (Elsevier/North-Holland).

Loughran, Tim, and Anand M. Vijh, 1997, Do long-term shareholders benefit from corporate acquisitions? *Journal of Finance* 52, 1765-1790.

Merton, Robert C., 1974, On the pricing of corporate debt: the risk structure of interest rates, Journal of Finance 29, 449-70.

Morck, Randall, Andrei Shleifer, and Robert W. Vishny, 1990, Do managerial objectives drive bad acquisitions? *Journal of Finance* 45, 31-48.

Mitchell, Karlyn and Walker, Mark D. 2008. "Bankers on Boards, Financial Constraints and Financial Distress," Working Paper.

Morck, R., Nakamura, M., 1999, Banks and Corporate Control in Japan, *Journal of Finance*, 54, 319-339.

Renneboog, Luc, and Peter G. Szilagyi, 2008, Corporate restructuring and bondholder wealth, *European Financial Management* 14, 792-819.

Rumble, Arienne S., and Joao A. C. Santos, 2006, The American keiretsu and universal banks: Investing, voting and sitting on nonfinancials' corporate boards, *Journal of Financial Economics* 80, 419-454.

Şişli-Ciamarra, Elif, 2011, Bankers on boards: Effects on financing decision, investment decision and performance, *Financial Management*, conditionally accepted.

Travlos, Nickolaos G., 1987, Corporate takeover bids, methods of payment, and bidding firms' stock returns, *Journal of Finance* 42, 943-963.

Tuch, Christian and Noel Sullivan, 2007, The Impact of Acquisitions on Firm Performance: A Review of the evidence, International Journal of Management Review, Vo. 9, Issue 2, pp. 141-170.

Wansley, James W., William R. Lane, and Ho C. Yang, 1987, Gains to bidder firms in cash and securities transactions, *Financial Review* 22, 403-414.

Wei, Chenyang, and David Yermack, 2010, Deferred Compensation, Risk, and Company Value: Investor Reactions to CEO Incentives, Working Paper.

#### Table 1: Banker presence on boards of directors

This table reports summary statistics of commercial bankers and investment bankers on boards of directors for the firm-level sample. We report firm-year observations for non-financial companies that are included in the S&P 500 Index at the beginning of 2002. For each year we report the percentage of firms with a commercial banker on the board, the percentage of companies with a creditor, those with an unaffiliated banker (and no creditor), as well as banks with an investment banker on their board. A creditor is an executive of a bank that has extended at least one loan to the company over the previous five years as a sole lender or a lead arranger in a syndicate. An unaffiliated banker is an executive of a bank with no outstanding loan relationship with the company during the previous five years. For ease of comparison with future tables, we report statistics for the regression sample that requires the availability of firm characteristics and governance variables.

Banker presence by fiscal year					
	Commercial banker	Creditor	Unaffiliated banker	Investment banker	
2002	32.6%	12.6%	20.0%	9.7%	
2003	32.6%	12.1%	20.5%	9.7%	
2004	31.6%	11.5%	20.2%	9.9%	
2005	24.9%	10.2%	14.7%	9.9%	
2006	24.9%	10.6%	14.2%	10.3%	
2007	26.1%	11.6%	14.5%	10.1%	
2002-2007	28.9%	11.4%	17.5%	10.0%	
Number of observation	ons: 2230				

Table 2: Banker presence during acquisitions

This table reports summary statistics of commercial bankers and investment bankers on boards of directors for the acquisition sample. The sample includes deals that were announced and completed between 2002 and 2007, where the acquirer is a non-financial company included in the S&P 500 at the beginning of 2002. We report the shares of acquisitions with presence of a commercial banker, creditor, unaffiliated banker (no creditor), and investment banker. For ease of comparison with future tables, we report statistics for the regression sample that requires the availability of market variables, as well as firm-level, deal-level, and governance controls.

	Number	Percentage
Commercial banker	488	29.7%
Creditor	258	15.7%
Unaffiliated banker	230	14.0%
Investment banker	165	10.1%
Number of observations:	1641	

Table 3: Creditor-director presence and firm-level characteristics

This table reports summary statistics of firm characteristics for firm-level sample (2002 to 2007). We report averages of firm-year observations for the full sample and for sub-samples with and without creditor presence on the board of directors. We also present p-values for comparison of means tests. Statistics are for the regression sample. Variable definitions are provided in the appendix. \* , \*\*, and \*\*\* denote significance at the 5%, 1%, and 0.1% level respectively.

	Full sample	Creditor present	Creditor not present	Hypothesis: Equal means (Prob>t)
Firm characteristics				
Size (\$ billion)	19.6	36.0	17.5	0.000***
Total debt / assets	0.24	0.29	0.23	0.000***
Cash / assets	0.13	0.11	0.14	0.002**
PPE / assets	0.30	0.33	0.30	0.027*
R&D / net sales	0.047	0.030	0.050	0.003**
ROA	0.15	0.14	0.15	0.081
Market-to-book ratio	1.74	1.51	1.77	0.001***
Stock return volatility	0.098	0.086	0.099	0.000***
CDS spread	98.7	76.6	102.2	0.143
Credit rating	8.1	7.6	8.2	0.008***
Default probability	0.0049	0.0051	0.0049	0.722
Short-term credit rtg. available	0.56	0.73	0.54	0.000***
Governance variables				
Board size	10.7	11.6	10.6	0.000***
Ratio of insiders	0.16	0.15	0.16	0.003**
Governance index	9.8	9.6	9.8	0.170
Number of observations: 2230				

Table 4: Creditor director presence and acquisition summary statistics

This table reports summary statistics of firm and acquisition characteristics for the acquisition sample (2002 to 2007). We report averages for the full sample and for sub-samples with and without creditor presence on the board of directors. We also present p-values for comparison of means tests. Statistics are for the regression sample. Variable definitions are provided in the appendix. \* , \*\*, and \*\*\* denote significance at the 5%, 1%, and 0.1% level respectively.

	Full sample	Creditor present	Creditor not present	Hypothesis: Equal means (Prob>t)
Acquirer financial characteristics				
Size (\$ billion)	30.9	27.1	31.6	0.303
Cash / assets	0.100	0.114	0.097	0.009**
Market-to-book ratio	1.68	1.50	1.71	0.001***
Total debt / assets	0.227	0.231	0.226	0.617
Stock return volatility	0.081	0.077	0.082	0.049*
Prior stock performance	0.137	0.158	0.134	0.205
CDS spread	57.5	49.3	59.1	0.227
Credit rating	6.8	6.7	6.8	0.439
Default probability	0.0040	0.0034	0.0042	0.156
Governance variables				
Board size	11.4	11.4	11.3	0.526
Ratio of insiders	0.146	0.137	0.147	0.058
Governance index	9.9	9.8	9.9	0.700
Deal attributes				
Diversifying acquisition	0.488	0.609	0.466	0.000***
Foreign target	0.316	0.364	0.307	0.071
Public target	0.133	0.128	0.135	0.776
Cash only payment	0.250	0.178	0.263	0.004**
Hostile takeover	0.0018	0.0000	0.0022	0.454
Number of observations: 1641				

Table 5: Creditor presence on board of directors and relative acquisition activity

The table reports summary statistics and regression results for acquisition activity. The sample includes firm-year observations (see Table 1) of the companies in our data set (2002 to 2007). A creditor is an executive of a bank that has extended at least one loan to the company over the previous five years as a sole lender or a lead arranger in a syndicate. Acquisition dummy is equal to 1 if there was an acquisition in the fiscal year, equal to 0 if there was not; Number of acquisitions counts acquisitions in the fiscal year; Acquisition value / total assets measures the relative value of acquisitions (assets are measured as of the end of the previous year); since acquisition size is not always available, the sample size (1604) is smaller for this variable. Panel A reports summary statistics. Panel B reports model estimation results; if there is statistically significant evidence for selection, we correct for it. Explanatory variables are defined in the appendix. Regressions include industry effects (1-digit SIC industry dummies) and year fixed effects. Absolute values of t-statistics (reported below coefficients) are calculated from standard errors that are robust and clustered by acquirer. \* , \*\*, and \*\*\* denote significance at the 5%, 1%, and 0.1% level respectively.

	Panel A: Comparison of means			
	Acquisition dummy	Number of acquisitions	Acquisition value / total assets	
All	0.475	1.109	0.029	
Creditor present	0.482	1.314	0.017	
Creditor not present	0.474	1.083	0.030	
Hypothesis: Equal means (Prob>t)	0.800	0.068	0.178	
	Panel B: Multivariate analy	sis		
	Acquisition dummy	Number of acquisitions	Acquisition value / total assets	
Creditor on board	0.0121 [0.917]	0.128 [0.569]	-0.00609 [0.407]	
Firm financial controls	[0.517]	[0.000]	[0.407]	
Size	0.149	0.379	-0.00585	
OIZO	[0.000]***	[0.000]***	[0.095]	
Market-to-book ratio	0.0977	0.193	0.0138	
manter to been raile	[0.008]**	[0.001]***	[0.036]*	
Cash / assets	-0.611	-0.936	0.0188	
	[0.063]	[0.070]	[0.566]	
Capital expenditures / assets	-2.170	-5.475	-0.0101	
	[0.066]	[0.000]***	[0.889]	
Total debt / assets	-0.847	-1.422	0.00544	
	[0.003]**	[0.000]***	[0.892]	
Governance controls				
Governance index	-0.0145	-0.0328	0.00177	
	[0.404]	[0.281]	[0.200]	
Ratio of insiders	0.0627	0.143	0.0234	
	[0.872]	[0.807]	[0.422]	
Constant	-1.210	-1.702	0.0365	
	[0.008]**	[0.008]**	[0.349]	
Observations	2230	2230	1604	
Method of estimation	Probit	Probit	Probit	
Self-selection	No	No	No	

Table 6: Creditor presence on board of directors and deal attributes

This table reports model estimation results for the acquisition sample (2002 to 2007). The dependent variables are: diversification dummy (equal to 1 if the 2-digit SIC codes for the acquirer and the target are different); foreign target dummy (equal to 1 if the target is a non-US company); public target dummy (defined similarly), and cash-only dummy (equal to 1 if the method of payment is identified as "cash-only" in the ThomsonOne database). The explanatory variables are defined as in the appendix. Regressions include industry effects (1-digit SIC industry dummies) and year fixed effects. Due to multi-collinearity the hostile takeover dummy is dropped in the public target model. Absolute values of t-statistics (reported below coefficients) are calculated from standard errors that are robust and clustered by acquirer. \* , \*\*, and \*\*\* denote significance at the 5%, 1%, and 0.1% level respectively.

	Diversifying acquisition	Foreign target	Public target	Cash-only payment
Creditor on board	0.261 [0.067]	0.106 [0.426]	0.185 [0.096]	-0.372 [0.001]***
Firm financial controls	[0.007]	[0.120]	[0.000]	[0.001]
Size	0.0594	0.0206	0.137	-0.123
Size	[0.387]	[0.703]	[0.014]*	[0.022]*
Market-to-book ratio	-0.121	-0.0354	0.0775	-0.0444
Warket-to-book ratio	[0.110]	[0.534]	[0.200]	[0.429]
Cash / assets	-1.103	0.170	0.748	1.176
Casii / assets	[0.095]	[0.740]	[0.171]	[0.068]
Total debt / assets	-1.202	-0.264	-0.463	-0.136
i otal debt / assets	[0.031]*	[0.546]	[0.302]	[0.739]
Prior stock performance	0.0556	-0.163	0.224	0.132
Thor stock periorillance	[0.693]	[0.220]	[0.207]	[0.302]
Stock return volatility	0.984	0.0327	-0.934	-1.040
Stock return volatility	[0.532]	[0.984]	[0.578]	[0.452]
Deal-specific controls	[0.332]	[0.904]	[0.576]	[0.432]
Diversifying acquisition		-0.199	-0.233	-0.0443
Diversifying acquisition		[0.019]*	[0.020]*	[0.624]
Foreign target	-0.198	[0.010]	-0.0664	-0.206
1 oroigh target	[0.019]*		[0.517]	[0.013]*
Public target	-0.245	-0.0855	[0.017]	1.299
r dono targot	[0.031]*	[0.458]		[0.000]***
Hostile takeover	1.139	0.288		-0.197
riostiic takeover	[0.011]*	[0.689]		[0.819]
Cash-only payment	-0.0468	-0.216	1.173	[0.010]
Cash only payment	[0.620]	[0.014]*	[0.000]***	
Governance controls	[0.020]	[0.017]	[0.000]	
Governance index	0.0444	-0.0231	0.0258	0.0183
Covornance index	[0.116]	[0.288]	[0.253]	[0.399]
Ratio of insiders	-0.0331	-0.365	-0.174	0.356
Tado of Holdoro	[0.957]	[0.452]	[0.773]	[0.498]
Constant	-1.138	0.166	-2.589	-0.231
Constant	[0.178]	[0.833]	[0.001]**	[0.745]
Observations	1641	1641	1638	1641
Method of estimation	Probit	Probit	Probit	Probit
Self-selection	No	No	No	No

#### Table 7: Creditor presence and equity and credit market reactions

This table reports equity, credit market, and firm value reactions around announcements of acquisitions completed by the companies in our sample (2002 to 2007). Market reactions are for the event window (t-5,t+5). Panel A reports means for three cases: all observations, creditor present on the board of directors, no creditor present. Equity return is the cumulative abnormal equity return; credit spread change is the percent change in the CDS spread, adjusted for contemporaneous market-wide CDS spread movements using the mean spread change; firm return is the abnormal firm return calculated using a weighted average of abnormal equity and excess debt return based on the return to a 5-year bond with a yield equal to Treasury yield plus CDS spread, weighted using market equity and book value of long-term debt; to control for outliers, spread changes and bond returns are winsorized at the 0.5% and the 99.5% levels. Panel B reports model estimation results; if there is statistically significant evidence for selection, we correct for it. The explanatory variables are defined as in the appendix. All regressions include industry effects (1-digit SIC industry dummies) and year fixed effects. P-values (reported below coefficients) are calculated from standard errors that are robust and clustered by year. \* , \*\*, and \*\*\* denote significance at the 5%, 1%, and 0.1% level respectively.

Panel A: Comparison of means						
Equity return CDS spread change Firm return						
All	-0.00131	0.00537	-0.00173			
	[0.280]	[0.096]	[0.229]			
Creditor present	0.00388	0.00242	0.00283			
	[0.216]	[0.563]	[0.269]			
Creditor not present	-0.00228	0.00592	-0.00258			
	[0.135]	[0.100]	[0.114]			
Hypothesis: Equal means (Prob>t)	0.00616	-0.00350	0.00540			
	[0.107]	[0.490]	[0.060]			
Number of observations: 1641						

Table 7 cont.

Panel B: Multivariate Analysis				
	Equity return	CDS spread change	Firm return	
Creditor on board	-0.0431	-0.0331	-0.0507	
	[0.049]*	[0.004]**	[0.000]***	
Firm financial controls				
Size	0.00301	0.00489	0.00292	
	[0.124]	[0.327]	[0.108]	
Market-to-book ratio	-0.00350	0.00581	-0.00435	
	[0.189]	[0.223]	[0.070]	
Cash / assets	-0.00948	-0.0647	-0.000815	
	[0.692]	[0.166]	[0.971]	
Total debt / assets	0.0455	-0.0229	0.0267	
	[0.085]	[0.508]	[0.063]	
Prior stock performance	0.000529	-0.0210	0.00291	
·	[0.936]	[0.057]	[0.601]	
Stock return volatility	0.0106	0.234	-0.0395	
,	[0.869]	[0.008]**	[0.506]	
Deal-specific controls	[ ]	(	1	
Diversifying acquisition	-0.00737	-0.0000519	-0.00637	
3 44 7	[0.124]	[0.994]	[0.128]	
Foreign target	0.00621	-0.00587	0.00544	
3 3 3 3 3 3	[0.077]	[0.340]	[0.101]	
Public target	-0.00186	0.00989	-0.00131	
	[0.613]	[0.000]***	[0.705]	
Hostile takeover	-0.00360	0.114	-0.00456	
	[0.798]	[0.167]	[0.713]	
Cash-only payment	0.0110	0.00334	0.00787	
cae c, payc	[0.002]**	[0.650]	[0.002]**	
Governance controls	[0.00=]	[0.000]	[0.00=]	
Governance index	0.000194	0.00208	0.000223	
	[0.770]	[0.069]	[0.701]	
Ratio of insiders	0.0295	-0.0682	0.0192	
	[0.224]	[0.009]**	[0.277]	
Constant	-0.0265	-0.0621	-0.0157	
Constant	[0.351]	[0.317]	[0.447]	
Observations	1641	1641	1641	
Method of estimation	Treatment Effects	Treatment Effects	Treatment Effects	
Self-selection	Yes	Yes	Yes	
_				
Test of indep. equations: P>chi <sup>2</sup>	0.004	0.000	0.000	

Table 8: Unaffiliated banker-director presence and acquisition summary statistics

This table reports summary statistics of firm characteristics for the acquisition sample (2002 to 2007). Summary statistics are for the full sample (unaffiliated banker or no banker), and for firm-years with an unaffiliated banker on the board or no banker on the board. Statistics are for the regression sample. The variables are defined in the appendix. \* , \*\*, and \*\*\* denote significance at the 5%, 1%, and 0.1% level respectively.

	Full sample	Unaffiliated banker present	No banker present	Hypothesis: Equal means (Prob>t)
Acquirer financial characteristics				
Size (\$ billion)	31,588	23,258	33,250	0.043*
Cash / assets	0.10	0.08	0.10	0.004**
Market-to-book ratio	1.71	1.67	1.72	0.410
Total debt / assets	0.23	0.22	0.23	0.499
Stock return volatility	0.082	0.078	0.083	0.150
Prior stock performance	0.13	0.16	0.13	0.094
CDS spread	61.7	41.5	65.7	0.028**
Credit rating	6.8	6.1	7.0	0.000***
Default probability	0.004	0.003	0.004	0.031*
Governance variables				
Board size	11.3	10.7	11.5	0.000***
Ratio of insiders	0.15	0.17	0.14	0.000***
Governance index	9.9	9.7	9.9	0.202
Deal attributes				
Diversifying acquisition	0.466	0.544	0.450	0.010**
Foreign target	0.307	0.344	0.300	0.193
Public target	0.134	0.152	0.131	0.390
Cash only payment	0.263	0.300	0.256	0.165
Hostile takeover	0.002	0.004	0.002	0.437
Number of observations: 1383				

Table 9: Unaffiliated banker-directors and wealth effects of acquisitions

This table reports regression results for equity, credit market, and firm value reactions around announcements of acquisitions for the sample with unaffiliated banker presence or no banker. Market reactions are for the event window (t-5,t+5) and are defined as in Table 7; to control for outliers, spread changes and bond returns are winsorized at the 0.5% and the 99.5% levels. Explanatory variables are defined in the appendix. All regressions include industry effects (1-digit SIC industry dummies) and year fixed effects. Absolute values of t-statistics (reported below coefficients) are calculated from standard errors that are robust and clustered by year. \*, \*\*, and \*\*\* denote significance at the 5%, 1%, and 0.1% level respectively.

	Equity return	CDS spread change	Firm return
Unaffiliated banker on board	0.00155	0.00260	0.00141
	[0.421]	[0.663]	[0.396]
Firm financial controls			
Size	0.00155	0.00260	0.00141
	[0.421]	[0.663]	[0.396]
Market-to-book ratio	-0.00174	0.00600	-0.00209
	[0.522]	[0.289]	[0.372]
Cash / assets	-0.0185	-0.0567	-0.0125
	[0.377]	[0.402]	[0.508]
Total debt / assets	0.0394	-0.0310	0.0244
	[0.185]	[0.428]	[0.152]
Prior stock performance	-0.000553	-0.0170	0.00327
	[0.942]	[0.236]	[0.638]
Stock return volatility	0.0121	0.202	-0.0244
•	[0.877]	[0.116]	[0.659]
Deal-specific controls			
Diversifying acquisition	-0.00770	-0.000538	-0.00672
	[0.137]	[0.942]	[0.145]
Foreign target	0.00621	-0.00375	0.00527
-	[0.127]	[0.642]	[0.142]
Public target	0.00125	0.00440	0.00163
_	[0.779]	[0.097]	[0.722]
Hostile takeover	-0.00430	0.114	-0.00267
	[0.754]	[0.234]	[0.832]
Cash-only payment	0.0101	0.00626	0.00688
	[0.033]*	[0.485]	[0.051]
Governance controls			
Governance index	-0.0000494	0.00159	-0.0000303
	[0.949]	[0.405]	[0.965]
Ratio of insiders	0.0385	-0.0763	0.0293
	[0.256]	[0.019]*	[0.244]
Constant	-0.0191	-0.0350	-0.0129
	[0.574]	[0.671]	[0.566]
Observations	1383	1383	1383
Method of estimation	OLS	OLS	OLS
Self-selection	No	No	No

Table 10: Investment bankers and wealth effects of acquisitions

This table reports model estimation results for equity, credit market, and firm value reactions around announcements of acquisitions for the sample with investment banker presence or no banker (same as in Table 9); if there is statistically significant evidence for selection, we correct for it. Market reactions are for the event window (t-5,t+5) and are defined as in Tables 7 and 9; to control for outliers, spread changes and bond returns are winsorized at the 0.5% and the 99.5% levels. Explanatory variables are defined in the appendix. All regressions include industry effects (1-digit SIC industry dummies) and year fixed effects. Absolute values of t-statistics (reported below coefficients) are calculated from standard errors that are robust and clustered by year. \*, \*\*, and \*\*\* denote significance at the 5%, 1%, and 0.1% level respectively.

	Equity return	CDS spread change	Firm return
Investment banker on board	-0.0565	-0.00347	-0.0370
	[0.004]**	[0.831]	[0.000]***
Firm financial controls			
Size	0.000854	0.00248	0.00100
	[0.637]	[0.669]	[0.506]
Market-to-book ratio	-0.00170	0.00575	-0.00214
	[0.597]	[0.314]	[0.389]
Cash / assets	-0.0282	-0.0601	-0.0180
	[0.151]	[0.349]	[0.231]
Total debt / assets	0.0274	-0.0331	0.0162
	[0.143]	[0.424]	[0.176]
Prior stock performance	-0.00118	-0.0167	0.00296
,	[0.880]	[0.253]	[0.645]
Stock return volatility	0.00446	0.191	-0.0326
,	[0.955]	[0.166]	[0.543]
Deal-specific controls	[]	[]	11
Diversifying acquisition	-0.00671	-0.0000965	-0.00622
, , ,	[0.091]	[0.990]	[0.090]
Foreign target	0.00712	-0.00329	0.00599
5 5	[0.047]*	[0.680]	[0.048]*
Public target	0.0000297	0.00463	0.000700
-	[0.994]	[0.105]	[0.858]
Hostile takeover	-0.00730	0.114	-0.00383
	[0.607]	[0.228]	[0.762]
Cash-only payment	0.00982	0.00642	0.00680
	[0.000]***	[0.472]	[0.004]**
Governance controls			
Governance index	0.0000525	0.00144	0.0000312
	[0.934]	[0.446]	[0.956]
Ratio of insiders	0.0273	-0.0716	0.0222
	[0.214]	[0.041]*	[0.207]
Constant	0.00111	-0.0289	0.000499
	[0.961]	[0.720]	[0.975]
Observations	1383	1383	1383
Method of estimation	Treatment Effects	OLS	Treatment Effects
Self-selection	Yes	No	Yes
Test of indep equations: P>chi2	0.000		0.000
rho	0.67		0.55